Author's response to reviews

Title: Effect of Mildly Elevated Thyroid-Stimulating Hormone during the First Trimester on Adverse Pregnancy Outcomes

Authors:
Ping Li (branda1982@163.com)
Shuo Lin (gzlinshuo@hotmail.com)
Ling Li (Linglizsy2000@163.com)
Jinhui Cui (Jinhuicui2000@163.com)
Shuisheng Zhou (Shuishengzhou2000@163.com)
Jianhui Fan (Jianhuifan2000@163.com)

Version: 3 Date: 01 Sep 2018

Author's response to reviews:

Thank you for your reviews. The responses are as follows:

1. English:

Response: Thank you for your comment. The whole manuscript had been reviewed by an english professional of Medjaden.

2. Figures:

Response: A separate heading of 'Figure Legends' had been added after the References and figure titles within the main text of the manuscript had been moved.